Author’s response to reviews

Title: PREVENTION PRACTICES INFLUENCING FREQUENCY OF OCCURRENCE OF VASO-OCCULSiVE CRISIS AMONG SICKLE CELL PATIENTS IN ABEOKUTA SOUTH LOCAL GOVERNMENT AREA OF OGUN STATE, NIGERIA.

Authors:

Olorunfemi Amoran (drfamoran@yahoo.com)
AHMED JIMOH (drjimoh@yahoo.com)
Omotola Ojo (dromotolajo@yahoo.com)
TEMITOPE KUPONIYI (drgb2002@yahoo.com)

Version: 3 Date: 14 Dec 2016

Author’s response to reviews:

Dear Editor

I have made several attempts to send this correction from the journal website without success before you now open the site for me. I have also sent it via e-mail to the editor several times. Sorry for any inconvenience caused.

Response to Comments and Editorial Requirements:

1) Please ensure that your paper is copy edited to improve clarity and understanding.

DONE

2) We understand you mention that written consent was obtained from the study participant and where the participants are minors, the consent was obtained from parents. Please confirm if it was written informed consent which was obtained from the parents/ guardians of the participants who are minors. Please ensure that a statement is added to reflect this is the paper.

DONE
3) Please ensure that a Declaration section is added to your paper along with all the subheadings as specified on the guidelines for the journal.

DONE

4) We noticed that your paper contains a large amount of text overlap, particularly in the Introduction and the Discussion. Please ensure that all text overlap are re-written. For further information, please visit the following page:
   http://www.biomedcentral.com/getpublished/editorial-policies

DONE

5) Please ensure that your paper is formatted according to the journal guidelines:
   http://bmchematol.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

DONE

Thank you for the opportunity to publish

Dr O.E. Amoran